

Rebuttal letter

Reviewer 02823559

1. Second aim has been removed.
2. Yes, we started enteral nutrition (via feeding jejunostomy) before hearing bowel sounds. We only observed time for return of bowel sounds.
3. Those 2 sentences in discussion have been removed.

Reviewer 00504545

Thanks for the appreciation. Tables have been rewritten as advised